Author’s response to reviews

Title: NEEDS ASSESSMENT OF EMERGENCY MEDICAL AND RESCUE SERVICES IN ABUJA/NIGERIA AND ENVIRONS

Authors:

ASAGA MAC PETER (pasaga123x@gmail.com)
Axel Kroeger (kroegera@who.int)
PHILOMENA AIRIOHUODION (ehidocng@gmail.com)

Version: 2 Date: 16 Apr 2019

Author’s response to reviews:

NEEDS ASSESSMENT OF EMERGENCY MEDICAL AND RESCUE SERVICES IN ABUJA/NIGERIA AND ENVIRONS" (EMMD-D-19-00026R1)

Authors Corrections

1. Line 45: NCDs (57.8%) and RTIs (42.8%) Highlighted in red in the main abstract body of the manuscript.

2. Line 110 (Now line 107). This study was conducted in Abuja federal capital territory and Keffi area council of Nasarawa state (Highlighted in red in the manuscript body).

3. Line 130-133 (Now 127-130). Measurements of delay times were carried out (the emergency call to attending the patient in the emergency ward) by the principal investigator, accompanying for 2 weeks the emergency team/staff of three institutions, observing their activities while rescuing victims and measuring the delay time between call and arrival at scene [Highlighted in red]

4. Line 144: Analysed [Highlighted in red in the manuscript body].

5. Line 256 (Now 260): Gout; ought to be analgesic or painkillers [Highlighted in red, Manuscript]


Results

7. [Tables]. It was an oversight, the Tables [1-4] are now included in the Body of the manuscript.

References:

8. All the references are inserted in the body of the manuscript where they ought to be; they are all highlighted in red.